



Limnology and Oceanography Letters 2023 © 2023 The Authors. Limnology and Oceanography Letters published by Wiley Periodicals LLC on behalf of Association for the Sciences of Limnology and Oceanography. doi: 10.1002/lol2.10368

LETTER

Combined exposure to CO₂ and H₂S significantly reduces the performance of the Mediterranean seagrass *Posidonia oceanica*: Evidence from a volcanic vent

Geraldina Signa,^{1,2} Valentina Sciutteri,³ Agostino Tomasello^{1,2} Valentina Costa^{1,4} Silvia Casabianca,^{2,5} Giovanna Cilluffo,^{1,2} Cristina Andolina,^{1,2} Salvatrice Vizzini^{1,2}

¹Department of Earth and Marine Sciences, University of Palermo, Palermo, Italy; ²Consorzio Nazionale Interuniversitario per le Scienze del Mare, Rome, Italy; ³Department of Integrative Marine Ecology (EMI), Stazione Zoologica Anton Dohrn, Messina, Italy; ⁴Department of Integrative Marine Ecology (EMI), Stazione Zoologica Anton Dohrn, Amendolara, Italy; ⁵Department of Biomolecular Sciences, Campus Enrico Mattei, University of Urbino, Urbino, Italy

Scientific Significance Statement

Seagrasses are expected to thrive in future acidified oceans due to their ability to overcome the low CO_2 diffusion into plant tissues. However, the effects of the co-occurrence of CO_2 and toxic compounds on the plant and, as a consequence, on the structure and function of the entire ecosystem are largely unknown. In this study, we show that the co-occurrence of CO_2 and H_2S in a shallow volcanic vent has detrimental effects on the seagrass *Posidonia oceanica*, from the leaf to the meadow level, due to sulfide intrusion, which impairs growth performance. The expected beneficial effects of high CO_2 levels on seagrasses may be dampened by other factors, highlighting the need to consider the natural complexity of ecosystems in ocean acidification studies.

Abstract

Although seagrasses are expected to thrive in future acidified oceans by overcoming low CO_2 diffusion into plant tissues, the co-occurrence of environmental stressors may affect their growth. Volcanic CO_2 vents are often associated with toxic gases and metal-rich fluids representing ideal sites to assess the effects of multiple stressors. We evaluated the response of *Posidonia oceanica* growing near shallow CO_2 vents characterized by H_2S spill-out by comparing meadow structure and phenology to an area with no gas emissions. Seagrass descriptors at meadow, shoot and leaf level indicated that *P. oceanica* previously described at CO_2 vents with no evidence of

Geraldina Signa and Valentina Sciutteri share co-first authorship.

^{*}Correspondence: agostino.tomasello@unipa.it

Associate editor: Barbara Robson

Author Contribution Statement: GS and VS compiled data, conducted the investigation process and prepared all figures. VS wrote the manuscript first draft. AT validated data. VC contributed to field surveys and data collection. GC applied statistical models to analyze data. SV developed the research question, designed the study, acquired project funding and resources and supervised the research. All authors reviewed and edited the article and approved the final version.

Additional Supporting Information may be found in the online version of this article.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

23782242, 0, Downloaded from https://aslopubs.onlinelibrary.wiley.com/doi/10.1002/b/12.10368 by University Degli Studi Di Pale, Wiley Online Library on [2705/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms

and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License



toxic inputs. Furthermore, the reduction in both leaf $\delta^{34}S$ and growth at the vent area indicates that sulfide intrusion occurs and affects seagrass growth performance, dampening the expected beneficial effects of high CO₂ levels.

Anthropogenic CO₂ emissions alter the carbon chemistry of the oceans by increasing pCO_2 and decreasing pH, a phenomenon known as ocean acidification (OA), which can alter the stability and functioning of marine ecosystems (Mostofa et al. 2016; Sunday et al. 2017; Cattano et al. 2018; Hall-Spencer and Harvey 2019; Aiuppa et al. 2021). Natural CO₂ vents are subtidal marine areas characterized by the emission of volcanic gases (mainly CO₂) that cause the formation of spatial pH gradients and create the conditions predicted by future acidification scenarios (i.e., pH decrease of 0.2/0.4 units by 2050-2100 according to the RCP8.5 scenario; Bindoff et al. 2022). They are therefore recognized as natural laboratories where scientific hypotheses on OA can be tested in situ (Hall-Spencer et al. 2008; Vizzini et al. 2017; Foo et al. 2018; Rastrick et al. 2018; Mirasole et al. 2020). Moreover, CO₂ vents are distributed across temperate and tropical latitudes overcoming the inherent limitations of laboratory and mesocosm studies by encompassing the environmental variability of natural systems. Although CO₂ accounts for the majority (90-99%) of volcanic emissions (Aiuppa et al. 2021), it is often associated with other gases (i.e., mainly N₂, O₂, CH₄, H₂S) and metals, which may represent non-negligible additional stressors and confounding factors when studying the ecological effects of OA (Vizzini et al. 2013; Agostini et al. 2015; Zitoun et al. 2020). Such a combination of chemical stressors may create extreme environmental conditions to which dwelling species must adapt in order to survive and thrive, and therefore provides opportunities to study their physiological, morphological, and behavioral responses and adaptations to future climate change conditions.

Seagrasses are habitat-forming species that play a crucial role in temperate and tropical coastal zones, supporting high biodiversity and delivering multiple ecosystem services (Fourqurean et al. 2012; Ondiviela et al. 2014). Despite some speciesspecificity (Russell et al. 2013; Ow et al. 2015), it is generally accepted that seagrasses, along with other primary producers, will be the "winners" in a high-CO2 world because of their strong affinity for dissolved inorganic carbon, mainly CO₂ and HCO₃, which can stimulate primary production leading to increased photosynthetic rates and standing crop (resourceeffect of CO₂) (Koch et al. 2013; Sunday et al. 2017). However, the combined exposure to high CO2 and environmental stressors such as toxicants (Vizzini et al. 2013), increased temperature (Repolho et al. 2017; Collier et al. 2018) or eutrophication (Campbell and Fourgurean 2014) may result in antagonistic effects where the stressor negative effects outweigh the beneficial effects of OA (Repolho et al. 2017; Perry et al. 2019). In contrast, to our knowledge, the effects of combined exposure to high CO_2 and H_2S , one of the most potent phytotoxins in the marine environment (Lamers et al. 2013),

have never been assessed. There is evidence that sediment porewater sulfides can enter and diffuse into seagrass tissues, disrupting plant physiology and exerting toxic effects. For instance, sulfide intrusion can suppress photosynthesis in *Zostera marina* (Holmer et al. 2005) and leaf production and growth in *Posidonia oceanica*, ultimately affecting survival (Calleja et al. 2007; Frederiksen et al. 2007; Garcias-Bonet et al. 2008).

In this context, the stable sulfur isotope ratio (δ^{34} S) is considered a valuable proxy for sulfide intrusion in seagrasses (Holmer and Hasler-Sheetal 2014). Potential sulfur sources for seagrasses have different isotopic signatures, with porewater sulfides being significantly negative (δ^{34} S = -10% to -20%) compared to seawater and porewater sulfates (δ^{34} S $\simeq +21$ and +20-60%, respectively) (Frederiksen et al. 2008). These differences can therefore be used to trace the origin of sulfur in seagrass tissues. Thus, a decrease of *P. oceanica* δ^{34} S from its normal values toward less positive values indicates that sulfur from sedimentary sulfides has accumulated in plant tissues and negative effects on plant performance are expected (Holmer and Hasler-Sheetal 2014).

Here we aimed to investigate the response of structural traits of P. oceanica, the foundation seagrass species forming one of the most important coastal ecosystems in the Mediterranean Sea, to combined exposure to CO₂ and H₂S in a natural hydrothermal vent, considering multiple levels of biological organization (i.e., meadow, shoot, and leaf) of the seagrass living near and away from the shallow vent. Although it was not possible to test the effects of CO₂ and H₂S individually, we hypothesized that the stressor effect of H₂S would dampen the resource-effect of CO₂, so that the expected increase in density, biomass, and canopy height at the meadows near the vent compared to a reference area would not occur. We also investigated the effects of co-exposure to CO₂ and H₂S on leaf epiphyte biomass and grazing pressure, to evaluate the propagation of the seagrass response throughout the meadow ecosystem.

Materials and methods

The study was carried out east of Panarea Island (Aeolian Archipelago, southern Tyrrhenian Sea) in the submerged area enclosed by the islets of Dattilo, Lisca Bianca, Bottaro, and Lisca Nera. Here, shallow vents dominated by CO_2 (~ 90–99 vol.%), but also featured by N_2 and H_2S emissions (both ~ 0–6 vol.%), and pH gradients (Italiano and Nuccio 1991; Steinbrückner 2009; Romano et al. 2019) characterize the subtidal environment. White/yellowish deposits have been recorded around volcanic emissions due to sulfur precipitation

and bacterial mat formation by sulfur-oxidizing bacteria (Gugliandolo et al. 2006; Steinbrückner 2009). The area has also been characterized by parossistic events, such as in November 2002, when explosive submarine activities west of Bottaro Islet severely affected the surrounding environment and biota (Aliani et al. 2010; Vizzini et al. 2010). The annual rhizome elongation of the nearby *P. oceanica* collapsed over the next 2 yr (Vizzini et al. 2010), followed by a gradual improvement with a complete recovery by 2010 (Noè et al. 2020).

Fieldwork was carried out at two areas with similar depth (~ 9 m), sandy bottom, wind, and currents, and characterized by the presence of *P. oceanica* meadows. A venting area, hereafter referred to as "vent," was identified east of Bottaro islet (38°38.270'N, 15°6.700'E), where continuous gas bubbling

with CO₂ (96–99%) and H₂S (0.5–2.5%) emissions was observed (Italiano and Nuccio 1991; Caracausi et al. 2005). A non-venting area, hereafter referred to as "reference," was located east of Lisca Nera (38°38.110'N, 15°6.460'E) and Dattilo (38°38.370'N, 15°5.980'E) islets, where gas emissions and fluid discharges were never recorded (Italiano and Nuccio 1991; Tassi et al. 2014). Within each area, two sites with similar environmental characteristics were randomly selected. The first sampling campaign, in July 2010, aimed to investigate the seagrass response at the meadow level only: shoot density was recorded by SCUBA divers at each site (six replicates) using a quadrat metal frame (40 × 40 cm) (Buia et al. 2004). During the second sampling campaign, in July 2011, the investigation at the meadow level was repeated and

Table 1. Mean \pm SD values of the *Posidonia oceanica* descriptors at the meadow (a) shoot (b), and leaf (c) levels measured at the vent and reference areas in 2010 and 2011. Leaf δ^{34} S is also shown (d).

Descriptor	Year	Area	
		Vent Mean ± SD	Reference Mean \pm SD
(a) Meadow			
Density (shoot m^{-2})	2010	$\textbf{261.5} \pm \textbf{58.8}$	435.9 ± 128.8
	2011	$\textbf{248.4} \pm \textbf{52.8}$	465.1 ± 104.4
(b) Shoots			
Total leaves (n) 2011		$\textbf{6.0} \pm \textbf{2.2}$	$\textbf{3.9}\pm\textbf{0.6}$
Adult leaves (<i>n</i>)		$\textbf{3.7}\pm\textbf{1.6}$	$\textbf{2.9}\pm\textbf{0.6}$
Intermediate leaves (n)		1.5 ± 0.5	1.0 ± 0.5
Juvenile leaves (n)		$\textbf{0.8} \pm \textbf{0.6}$	0.0
Surface (cm ² shoot ⁻¹)		61.5 ± 23.7	243.0 ± 73.8
Biomass (mg dw shoot $^{-1}$)		$\textbf{345.4} \pm \textbf{146.8}$	1350.4 ± 476.6
Green tissue (%)		$\textbf{0.9} \pm \textbf{0.0}$	1.0 ± 0.0
Brown tissue (%)		$\textbf{0.1}\pm\textbf{0.0}$	$\textbf{0.0} \pm \textbf{0.0}$
Leaf epiphyte biomass (mg dw shoot ⁻¹)		$\textbf{25.6} \pm \textbf{11.3}$	172.0 ± 98.0
Coefficient A (%)		0.4 ± 0.2	$\textbf{0.7}\pm\textbf{0.2}$
(c) Leaves			
Adult leaves			
Length (cm) 2011		$\textbf{22.7} \pm \textbf{6.4}$	$\textbf{79.7} \pm \textbf{23.7}$
Width (cm)		0.6 ± 0.1	$\textbf{0.8}\pm\textbf{0.1}$
Biomass (mg dw leaf ⁻¹)		$\textbf{79.0} \pm \textbf{28.2}$	$\textbf{382.4} \pm \textbf{134.5}$
Sheath length (cm)		1.9 ± 0.6	$\textbf{4.4} \pm \textbf{1.1}$
Sheath biomass (mg dw leaf ⁻¹)		15.5 ± 11.3	$\textbf{45.8} \pm \textbf{27.1}$
Epiphyte biomass (mg dw leaf ⁻¹)		$\textbf{6.3} \pm \textbf{3.8}$	53.0 ± 30.4
Intermediate leaves			
Length (cm)		14.8 ± 8.7	68.2 ± 26.7
Width (cm)		0.6 ± 0.1	$\textbf{0.8}\pm\textbf{0.1}$
Biomass (mg dw leaf ⁻¹)		$\textbf{35.4} \pm \textbf{22.1}$	241.5 ± 122.9
Epiphyte biomass (mg dw leaf ⁻¹)		1.5 ± 2.4	18.3 ± 14.7
(d) Stable isotopes			
Leaf δ ³⁴ S (‰) 2011		10.6 ± 0.8	15.5 ± 0.5

extended to the shoot and leaf level: five shoots were randomly collected at each site for the assessment of shoot and leaf descriptors (Table 1, Buia et al. 2004) including the analysis of stable sulfur isotopes (δ^{34} S) of *P. oceanica* leaves and the associated epiphyte biomass. At both sampling campaigns, temperature, salinity, dissolved oxygen, and pH at the sediment–seawater interface were recorded in triplicate at each site using an HYDROLAB DS5 multiparametric probe.

In the laboratory, after leaf biometry estimation, freezedried leaves (both adult and intermediate) from each *P. oceanica* shoot were ground, homogenized, and packed into tin capsules for stable sulfur isotope analysis. Samples were then analyzed using a Thermo IRMS (Delta V ADVANTAGE) coupled with an elemental analyzer (EA Flash 2000). Results are given in the δ notation as per mil deviation from the international standard (V-CDT) as follows: $\delta^{34}S = [(^{34}S/^{32}S_{sample})/(^{34}S/^{32}S_{standard}) - 1] \times 10^3$.

The analytical precision, based on the standard deviation of replicates of internal standards (IAEA-SO-6, IAEA-S-1, NBS 127) was \pm 0.2‰.

Seawater variables and meadow descriptors were analyzed using the following design: Area (two levels: vent, reference) and Year (two levels: 2010, 2011) as orthogonal and fixed factors, and Site as a random factor (nested in both Area and Year). Shoot and leaf descriptors, only available for 2011, were tested considering only the factors Area and Site (nested in Area). Univariate ANOVA was performed on seawater and seagrass descriptors at all levels (meadow, shoot, leaf) and on $\delta^{34}S$ (STATISTICA 12, StatSoft). The normality assumption was



Fig. 1. Density of *Posidonia oceanica* at the two sites of the vent and reference areas in 2010 and 2011 (n = 12). Whiskers indicate the non-outlier range of variation; boxes: 25th to 75th percentiles. Significant differences between areas are indicated (*see* Supporting Information Table S2 for details).

tested using the Shapiro–Wilk test and the log(x + 1) transformation was applied when the normality assumption did not hold. Homogeneity of variances was tested using Cochran's test. Multivariate permutational ANOVA (PERMANOVA; PRIMER 6 and PERMANOVA+ package) was run on the seagrass descriptors grouped by level of organization. PER-MANOVA was run on the Euclidean distance matrices obtained from log(x + 1) transformed normalized data, and the Monte-Carlo test was run in the case of low permutations (< 100). Pairwise post hoc tests were conducted when significant effects of the interactions of the factors Area and Year were found.

Results

Seawater variables showed similar values between areas, years, and sites except for pH, which was significantly lower at the vent than at the reference area (Supporting Information Table S1; Signa et al. 2023). At the meadow level, the density of *P. oceanica* was significantly lower at the vent than at the reference area, with no inter-annual or inter-site variability (Fig. 1; Table 1; Supporting Information Table S2).

At the shoot level, significant differences were detected between areas and sites (Supporting Information Table S2). As shown in Table 1, shoots at the vent area were characterized by a higher number of leaves for all age categories (i.e., adult, intermediate and juvenile leaves) together with a higher percentage of brown tissue than at the reference area. On the other hand, shoots showed higher leaf surface and biomass, epiphyte biomass, percentage of green tissue and eroded apices (Coefficient A) in the reference area (Table 1; Supporting Information Table S2). There were no juvenile leaves in the shoots at the reference area (Table 1).

At the leaf level, the differences between the two areas became more pronounced, while the inter-site variability disappeared. Specifically, the values of all descriptors of both adult and intermediate leaves were lower at the vent area than at the reference area (Table 1; Supporting Information Table S2). Similarly, δ^{34} S values of *P. oceanica* leaves were significantly lower at the vent than at the reference area, with no difference between sites (Table 1; Supporting Information Table S2).

Discussion

This study showed significant impairment in meadow structure and plant phenology of *P. oceanica* living near a shallow volcanic vent affected by both CO_2 and H_2S inputs compared to a reference area. Although the peculiarities of the vent area did not allow to investigate separately the effects of CO_2 and toxic H_2S emissions, the changes in seagrass traits (i.e., reduced density, biomass, canopy height and leaf size), together with the evidence of sulfide intrusion, suggest that CO_2 and H_2S may have acted antagonistically, dampening the

expected CO_2 -resource positive effect and resulting in an overall negative response of the plant at the vent area.

Hydrogen sulfide concentration in the vent area was not measured in this study, but the presence of white mat-forming bacteria alongside the typical sulfur smell confirms the presence of this toxic gas, as previously evidenced (Italiano and Nuccio 1991; Caracausi et al. 2005). Furthermore, the decrease in δ^{34} S of *P. oceanica* leaves from the vent area compared to the reference area is a valuable indication not only of the presence of sulfides in the former (Italiano and Nuccio 1991; Steinbrückner 2009; Romano et al. 2019), but also of sulfide intrusion into plant tissues (Holmer and Hasler-Sheetal 2014). Hydrothermal and biogenic porewater sulfides, which are abundant in Panarea sediments (Peters et al. 2011), are thought to be an important source of sulfur in leaf tissues near vent sites, as they enter through roots and are transferred to rhizomes and leaves, when present (Frederiksen et al. 2008). As porewater sulfides typically have negative δ^{34} S values, a decrease in δ^{34} S in seagrass tissues is a valuable proxy for sulfide exposure and intrusion (Holmer and Hasler-Sheetal 2014). Indeed, δ^{34} S of P. oceanica leaves from the reference area falls within the range (+15-27‰) reported by Holmer and Hasler-Sheetal (2014) for areas unaffected by sulfides, while the decrease of about 5% found near the Panarea vents provides evidence that some amount of sulfur from sedimentary sulfides has accumulated in *P. oceanica* leaves. Furthermore, the reduction in both leaf δ^{34} S and biometric parameters provides clear evidence of the stressor effect of sulfide intrusion on seagrass performance in line with previous studies showing reduced growth, meristem activity and above-ground productivity and increased mortality following sulfide intrusion in P. oceanica (Calleja et al. 2007; Frederiksen et al. 2008; Garcias-Bonet et al. 2008).

The reduction in shoot density and biomass found here was opposite to previous findings on P. oceanica living near the volcanic vent of Ischia (Italy, Tyrrhenian Sea) where volcanic gas emissions are constituted by CO2 with no hydrogen sulfide (Foo et al. 2018). Under these "pure" acidic conditions, P. oceanica thrives at densities twice as high as those observed in the reference area and 10-fold higher than those found in the vent area of the present study (Hall-Spencer et al. 2008; Mecca et al. 2020; Mirasole et al. 2021). In contrast, patterns similar to those found in this study were highlighted for Cymodocea nodosa growing nearby Vulcano vents (Eolian Archipelago) (Apostolaki et al. 2014), where the seagrass is likely to be stressed by H_2S_1 , CH₄, and metals (Capaccioni et al. 2001; Vizzini et al. 2013). The contrasting characteristics of gas emissions at Panarea and Ischia vents may also account for the different expression patterns of key stress-related genes previously observed in P. oceanica near the two vent systems, representing the result of site-specific environmental stress rather than the reflection of contrasting homeostatic evolutionary compensation (Lauritano et al. 2015).

As regards plant phenology, it is worth noting that the leaves of the shoots at the vent area were more numerous than at reference conditions, but they were also shorter and thinner, with a higher proportion of brown than green tissue. Changes in leaf color may be the result of (i) the alteration of the ratio between phaeopigments and chlorophyll *a*, as already observed in *Zostera noltii* under thermal stress (i.e., warming) (Repolho et al. 2017) and/or (ii) a faster leaf aging process under sulfide stress (Lamers et al. 2013). This is further supported by the presence of young leaves only in plants close to the vents, consistent with increased leaf turnover (Perry et al. 2019) probably as a physiological mechanism to offset the early leaf aging.

The undersizing of P. oceanica leaves in response to extreme environmental conditions has already been observed in Mediterranean marine areas (Gravili et al. 2021; Mancuso et al. 2023; Nguyen et al. 2023), including Panarea, where P. oceanica "bonsai" shoots have recently been described (Gambi et al. 2023). Consistently, we found smaller plants near the vents, but with more leaves than at reference sites, confirming the complexity of the plant response to multiple gases and the need for further investigation, especially to distinguish acclimatization mechanisms from young plants recovering after strong disturbances (parossistic events). The divergent pattern found here between high leaf abundance and turnover and low leaf size at the vent area seems to mirror the occurrence of acclimatization mechanisms in P. oceanica exposed to combined CO2 and H2S emissions, resulting in a scarce efficiency of the plant to allocate the additional carbon of volcanic origin into new biomass under stressful conditions (Invers et al. 2001; Apostolaki et al. 2014; Vizzini et al. 2019), rather than a higher grazing pressure. Indeed, the low Coefficient A (i.e., number of eroded leaf apices) found in the vents confirms this hypothesis, suggesting low exploitation of P. oceanica by herbivores and macrograzers, contrary to previous observations in the "pure" CO2 Ischia vents (Donnarumma et al. 2014; Mirasole et al. 2020). Indeed, OA is commonly associated with increased nutritional value (low C : N ratio) and decreased deterrent compounds (i.e., phenolics) in P. oceanica and, consequently, higher seagrass exploitation by herbivores (Ricevuto et al. 2015), all aspects that deserve further research under the complex environmental conditions of the Panarea vent system.

Differently, the drastic reduction in epiphyte biomass observed near the vents is consistent with the classic response of calcifying epibionts to OA (Hall-Spencer et al. 2008; Nogueira et al. 2017), although the low leaf size and faster leaf turnover may have certainly played a role, resulting in a reduced leaf surface and time available for epiphyte colonization. The low epiphyte colonization observed in the vent area may also have increased light exposure, which in turn may have contributed to the previously discussed leaf aging process and die-off (brown tissue) (Munné-Bosch and Alegre 2002; Zimmermann and Zentgraf 2005).

Overall, the outcomes of this study underlined highly stressful conditions associated with sulfide intrusion in the tissues of *P. oceanica* living near the Panarea vents and the consequent negative response in plant density and biomass. This response contrasts with the thriving condition of the same

species previously described at the Ischia vents, where there is no evidence of toxic inputs (Hall-Spencer et al. 2008; Mecca et al. 2020; Mirasole et al. 2021), suggesting that H₂S emissions from volcanic vents, when they occur, affect seagrass performance, net of the potential beneficial effects of increasing CO2 concentrations. Indeed, we observed here that the response of seagrasses to OA can deviate from expectations when increased CO₂ is combined with chemical stressors, and that this response is as complex as highly context-dependent, given the variability of volcanic input at shallow vents (Koch et al. 2013; Mostofa et al. 2016; Sunday et al. 2017; Collier et al. 2018; Hall-Spencer and Harvey 2019; Perry et al. 2019). Moreover, we revealed that eco-physiological mechanisms of acclimatization occur at multiple levels (from leaf to meadow) of P. oceanica long-term exposed to CO₂ and toxicants, paving the way for further research on the response of seagrass ecosystems under future global change scenarios. Although the results of this study suggest that the combination of CO₂ and H₂S may act antagonistically on *P. oceanica*, further studies are needed to determine the specific underlying mechanisms, using specific multiple stressor designs and possibly combining field and manipulative approaches.

Data availability statement

Data and metadata are available on Zenodo at https://zenodo.org/records/8425339.

References

- Agostini, S., and others. 2015. Geochemistry of two shallow CO₂ seeps in Shikine Island (Japan) and their potential for ocean acidification research. Reg. Stud. Mar. Sci. **2**: 45–53. doi:10.1016/j.rsma.2015.07.004
- Aiuppa, A., J. M. Hall-Spencer, M. Milazzo, G. Turco, S. Caliro, and R. Di Napoli. 2021. Volcanic CO₂ seep geochemistry and use in understanding ocean acidification. Biogeochemistry **152**: 93–115. doi:10.1007/s10533-020-00737-9
- Aliani, S., G. Bortoluzzi, G. Caramanna, and F. Raffa. 2010. Seawater dynamics and environmental settings after November 2002 gas eruption off Bottaro (Panarea, Aeolian Islands, Mediterranean Sea). Cont. Shelf Res. **30**: 1338– 1348. doi:10.1016/j.csr.2010.04.016
- Apostolaki, E. T., S. Vizzini, I. E. Hendriks, and Y. S. Olsen. 2014. Seagrass ecosystem response to long-term high CO₂ in a Mediterranean volcanic vent. Mar. Environ. Res. **99**: 9–15. doi:10.1016/j.marenvres.2014.05.008
- Bindoff, N. L., and others. 2022. Changing ocean, marine ecosystems, and dependent communities, p. 447–588. In IPCC special report on the ocean and cryosphere in a changing climate. Cambridge University Press. doi:10.1017/978100 9157964.007
- Buia, M. C., M. C. Gambi, and M. Dappiano. 2004. The seagrass systems, p. 133–184. *In Mediterranean marine ben-thos: A manual of methods for its sampling and study*. Biologia Marina Mediterranea.

- Calleja, M. L., N. Marbà, and C. M. Duarte. 2007. The relationship between seagrass (*Posidonia oceanica*) decline and sulfide porewater concentration in carbonate sediments. Estuar. Coast. Shelf Sci. **73**: 583–588. doi:10.1016/j.ecss. 2007.02.016
- Campbell, J. E., and J. W. Fourqurean. 2014. Ocean acidification outweighs nutrient effects in structuring seagrass epiphyte communities. J. Ecol. **102**: 730–737. doi:10.1111/ 1365-2745.12233
- Capaccioni, B., F. Tassi, and O. Vaselli. 2001. Organic and inorganic geochemistry of low temperature gas discharges at the Baia di Levante beach, Vulcano Island, Italy. J. Volcanol. Geotherm. Res. **108**: 173–185. doi:10.1016/ S0377-0273(00)00284-5
- Caracausi, A., M. Ditta, F. Italiano, M. Longo, P. M. Nuccio, A. Paonita, and A. Rizzo. 2005. Changes in fluid geochemistry and physico-chemical conditions of geothermal systems caused by magmatic input: The recent abrupt outgassing off the Island of Panarea (Aeolian Islands, Italy). Geochim. Cosmochim. Acta **69**: 3045–3059. doi:10.1016/j.gca.2005. 02.011
- Cattano, C., J. Claudet, P. Domenici, and M. Milazzo. 2018. Living in a high CO₂ world: A global meta-analysis shows multiple trait-mediated fish responses to ocean acidification. Ecological monographs 88: 320–335. doi:10.1002/ ecm.1297
- Collier, C. J., L. Langlois, Y. Ow, C. Johansson, M. Giammusso, M. P. Adams, K. R. O'Brien, and S. Uthicke. 2018. Losing a winner: Thermal stress and local pressures outweigh the positive effects of ocean acidification for tropical seagrasses. New Phytol. 219: 1005–1017. doi:10.1111/nph.15234
- Donnarumma, L., C. Lombardi, S. Cocito, and M. C. Gambi. 2014. Settlement pattern of *Posidonia oceanica* epibionts along a gradient of ocean acidification: An approach with mimics. Mediterr. Mar. Sci. **15**: 498–509. doi:10.12681/ mms.677
- Foo, S. A., M. Byrne, E. Ricevuto, and M. C. Gambi. 2018. The carbon dioxide vents of Ischia, Italy, a natural system to assess impacts of ocean acidification on marine ecosystems: An overview of research and comparisons with other vent systems. Oceanogr. Mar. Biol. 56: 237–310. doi:10. 1201/9780429454455-4
- Fourqurean, J. W., and others. 2012. Seagrass ecosystems as a globally significant carbon stock. Nat. Geosci. **5**: 505–509. doi:10.1038/ngeo1477
- Frederiksen, M. S., M. Holmer, E. Díaz-Almela, N. Marba, and C. M. Duarte. 2007. Sulfide invasion in the seagrass *Posidonia oceanica* at Mediterranean fish farms: Assessment using stable sulfur isotopes. Mar. Ecol. Prog. Ser. **345**: 93– 104. doi:10.3354/meps06990
- Frederiksen, M. S., M. Holmer, M. Pérez, O. Invers, J. M. Ruiz, and B. B. Knudsen. 2008. Effect of increased sediment sulfide concentrations on the composition of stable sulfur

isotopes (δ^{34} S) and sulfur accumulation in the seagrasses *Zostera marina* and *Posidonia oceanica*. J. Exp. Mar. Biol. Ecol. **358**: 98–109. doi:10.1016/j.jembe.2008.01.021

- Gambi, M. C., V. Esposito, and L. Marín-Guirao. 2023. *Posidonia bonsai*: Dwarf *Posidonia oceanica* shoots associated to hydrothemal vent systems (Panarea Island, Italy). Aquat. Bot. **185**: 103611. doi:10.1016/j.aquabot.2022.103611
- Garcias-Bonet, N., N. Marbà, M. Holmer, and C. M. Duarte. 2008. Effects of sediment sulfides on seagrass *Posidonia* oceanica meristematic activity. Mar. Ecol. Prog. Ser. **372**: 1–6. doi:10.3354/meps07714
- Gravili, C., F. Cozzoli, and M. C. Gambi. 2021. Epiphytic hydroids on *Posidonia oceanica* seagrass meadows are winner organisms under future ocean acidification conditions: Evidence from a CO_2 vent system (Ischia Island, Italy). Eur. Zool. J. **88**: 472–486. doi:10.1080/24750263.2021.1899317
- Gugliandolo, C., F. Italiano, and T. L. Maugeri. 2006. The submarine hydrothermal system of Panarea (southern Italy): Biogeochemical processes at the thermal fluids-sea bottom interface. Ann. Geophys. 49: 783–792. doi:10.4401/ ag-3139
- Hall-Spencer, J. M., and B. P. Harvey. 2019. Ocean acidification impacts on coastal ecosystem services due to habitat degradation. Emerg. Top. Life Sci. 3: 197–206. doi:10.1042/ ETLS20180117
- Hall-Spencer, J. M., and others. 2008. Volcanic carbon dioxide vents show ecosystem effects of ocean acidification. Nature 454: 96–99. doi:10.1038/nature07051
- Holmer, M., M. S. Frederiksen, and H. Møllegaard. 2005. Sulfur accumulation in eelgrass (*Zostera marina*) and effect of sulfur on eelgrass growth. Aquat. Bot. **81**: 367–379. doi:10. 1016/j.aquabot.2004.12.006
- Holmer, M., and H. Hasler-Sheetal. 2014. Sulfide intrusion in seagrasses assessed by stable sulfur isotopes—A synthesis of current results. Front. Mar. Sci. 1: 1–12. doi:10.3389/fmars. 2014.00064
- Invers, O., R. C. Zimmerman, R. S. Alberte, M. Pérez, and J. Romero. 2001. Inorganic carbon sources for seagrass photosynthesis: An experimental evaluation of bicarbonate use in species inhabiting temperate waters. J. Exp. Mar. Biol. Ecol. 265: 203–217. doi:10.1016/S0022-0981(01) 00332-X
- Italiano, F., and P. M. Nuccio. 1991. Geochemical investigations of submarine volcanic exhalations to the east of Panarea, Aeolian Islands, Italy. J. Volcanol. Geotherm. Res. 46: 125–141. doi:10.1016/0377-0273(91)90079-F
- Koch, M., G. Bowes, C. Ross, and X. H. Zhang. 2013. Climate change and ocean acidification effects on seagrasses and marine macroalgae. Glob. Chang. Biol. 19: 103–132. doi: 10.1111/j.1365-2486.2012.02791.x
- Lamers, L. P. M., and others. 2013. Sulfide as a soil phytotoxin—A review. Front. Plant Sci. **4**: 1–14. doi:10. 3389/fpls.2013.00268

- Lauritano, C., and others. 2015. Response of key stress-related genes of the seagrass *Posidonia oceanica* in the vicinity of submarine volcanic vents. Biogeosciences **12**: 4185–4195. doi:10.5194/bg-12-4185-2015
- Mancuso, F. P., J. Bernardeau-Esteller, M. Spinelli, G. Sarà, J. M. Ruiz, S. Calvo, and A. Tomasello. 2023. Life on the edge: Adaptations of *Posidonia oceanica* to hypersaline conditions in a Mediterranean lagoon system. Environ. Exp. Bot. **210**: 105320. doi:10.1016/j.envexpbot.2023.105320
- Mecca, S., E. Casoli, G. Ardizzone, and M. C. Gambi. 2020. Effects of ocean acidification on phenology and epiphytes of the seagrass *Posidonia oceanica* at two CO₂ vent systems of Ischia (Italy). Mediterr. Mar. Sci. **21**: 70. doi:10.12681/mms.20795
- Mirasole, A., F. Badalamenti, A. Di Franco, M. C. Gambi, and N. Teixidó. 2021. Boosted fish abundance associated with *Posidonia* oceanica meadows in temperate shallow CO₂ vents. Sci. Total Environ. **771**: 145438. doi:10.1016/j.scitotenv.2021.145438
- Mirasole, A., G. Signa, P. Gianguzza, C. Bonaviri, A. Mazzola, and S. Vizzini. 2020. Fish assemblages cope with ocean acidification in a shallow volcanic CO_2 vent benefiting from an adjacent recovery area. Mar. Environ. Res. **157**: 104851. doi:10.1016/j.marenvres.2019.104851
- Mostofa, K. M. G., and others. 2016. Reviews and syntheses: Ocean acidification and its potential impacts on marine ecosystems. Biogeosciences **13**: 1767–1786. doi:10.5194/ bg-13-1767-2016
- Munné-Bosch, S., and L. Alegre. 2002. Plant aging increases oxidative stress in chloroplasts. Planta **214**: 608–615. doi: 10.1007/s004250100646
- Nguyen, H. M., and others. 2023. Signs of local adaptation by genetic selection and isolation promoted by extreme temperature and salinity in the Mediterranean seagrass *Posidonia oceanica*. Mol. Ecol. **32**: 4313–4328. doi:10.1111/mec.17032
- Noè, S., C. Bellavia, S. Calvo, A. Mazzola, M. Pirrotta, M. Sciandra, S. Vizzini, and A. Tomasello. 2020. Resilience of the seagrass *Posidonia oceanica* following pulse-type disturbance. Mar. Environ. Res. **159**: 105011. doi:10.1016/j.mar envres.2020.105011
- Nogueira, P., M. C. Gambi, S. Vizzini, G. Califano, A. M. Tavares, R. Santos, and B. Martínez-Crego. 2017. Altered epiphyte community and sea urchin diet in *Posidonia oceanica* meadows in the vicinity of volcanic CO₂ vents. Mar. Environ. Res. **127**: 102–111. doi:10.1016/j.marenvres.2017.04.002
- Ondiviela, B., I. J. Losada, J. L. Lara, M. Maza, C. Galván, T. J. Bouma, and J. van Belzen. 2014. The role of seagrasses in coastal protection in a changing climate. Coast. Eng. **87**: 158–168. doi:10.1016/j.coastaleng.2013.11.005
- Ow, Y. X., C. J. Collier, and S. Uthicke. 2015. Responses of three tropical seagrass species to CO₂ enrichment. Mar. Biol. **162**: 1005–1017. doi:10.1007/s00227-015-2644-6
- Perry, D., and others. 2019. Global environmental changes negatively impact temperate seagrass ecosystems. Ecosphere **10**: e02986. doi:10.1002/ecs2.2986

- Peters, M., H. Strauss, S. Petersen, N. A. Kummer, and C. Thomazo. 2011. Hydrothermalism in the Tyrrhenian Sea: Inorganic and microbial sulfur cycling as revealed by geochemical and multiple sulfur isotope data. Chem. Geol. 280: 217–231. doi:10.1016/j.chemgeo.2010.11.011
- Rastrick, S. S. P., and others. 2018. Using natural analogues to investigate the effects of climate change and ocean acidification on Northern ecosystems. ICES J. Mar. Sci. **75**: 2299– 2311. doi:10.1093/icesjms/fsy128
- Repolho, T., and others. 2017. Seagrass ecophysiological performance under ocean warming and acidification. Sci. Rep. 7: 41443. doi:10.1038/srep41443
- Ricevuto, E., S. Vizzini, and M. C. Gambi. 2015. Ocean acidification effects on stable isotope signatures and trophic interactions of polychaete consumers and organic matter sources at a CO₂ shallow vent system. J. Exp. Mar. Biol. Ecol. **468**: 105–117. doi:10.1016/j.jembe.2015.03.016
- Romano, D., A. Gattuso, M. Longo, C. Caruso, G. Lazzaro, A. Corbo, and F. Italiano. 2019. Hazard scenarios related to submarine volcanic-hydrothermal activity and advanced monitoring strategies: A study case from the Panarea volcanic group (Aeolian Islands, Italy). Geofluids **2019**: 1–15. doi:10.1155/2019/8728720
- Russell, B. D., S. D. Connell, S. Uthicke, N. Muehllehner, K. E. Fabricius, and J. M. Hall-Spencer. 2013. Future seagrass beds: Can increased productivity lead to increased carbon storage? Mar. Pollut. Bull. **73**: 463–469. doi:10.1016/j.mar polbul.2013.01.031
- Signa, G., V. Sciutteri, A. Tomasello, V. Costa, S. Casabianca, G. Cilluffo, C. Andolina, and S. Vizzini. 2023. Combined exposure to CO₂ and H₂S significantly reduces the performance of the Mediterranean seagrass *Posidonia oceanica*: Evidence from a volcanic CO₂ vent. Zenodo. doi:10.5281/zenodo.8425339
- Steinbrückner, D. 2009. Quantification of submarine degassing of Panarea volcano in the Aeolian archipelago, Italy. FOG Freib. Online Geosci. 23: 1–113. doi:10.23689/ fidgeo-886
- Sunday, J. M., and others. 2017. Ocean acidification can mediate biodiversity shifts by changing biogenic habitat. Nat. Clim. Change 7: 81–85. doi:10.1038/nclimate3161
- Tassi, F., B. Capaccioni, and O. Vaselli. 2014. Compositional spatial zonation and 2005-2013 temporal evolution of the hydrothermal-magmatic fluids from the submarine fumarolic field at Panarea Island (Aeolian Archipelago, southern Italy). J. Volcanol. Geotherm. Res. 277: 41–50. doi:10. 1016/j.jvolgeores.2014.03.010

- Vizzini, S., E. T. Apostolaki, E. Ricevuto, P. Polymenakou, and A. Mazzola. 2019. Plant and sediment properties in seagrass meadows from two Mediterranean CO₂ vents: Implications for carbon storage capacity of acidified oceans. Mar. Environ. Res. **146**: 101–108. doi:10.1016/j.marenvres.2019.03.001
- Vizzini, S., R. Di Leonardo, V. Costa, C. D. Tramati, F. Luzzu, and A. Mazzola. 2013. Trace element bias in the use of CO₂ vents as analogues for low pH environments: Implications for contamination levels in acidified oceans. Estuar. Coast. Shelf Sci. **134**: 19–30. doi:10.1016/j.ecss.2013.09.015
- Vizzini, S., B. Martínez-Crego, C. Andolina, A. Massa-Gallucci, S. D. Connell, and M. C. Gambi. 2017. Ocean acidification as a driver of community simplification via the collapse of higher-order and rise of lower-order consumers. Sci. Rep. 7: 4018. doi:10.1038/s41598-017-03802-w
- Vizzini, S., and others. 2010. Effect of explosive shallow hydrothermal vents on δ^{13} C and growth performance in the seagrass *Posidonia oceanica*. J. Ecol. **98**: 1284–1291. doi: 10.1111/j.1365-2745.2010.01730.x
- Zimmermann, P., and U. Zentgraf. 2005. The correlation between oxidative stress and leaf senescence during plant development. Cell. Mol. Biol. Lett. **10**: 515–534.
- Zitoun, R., and others. 2020. A unique temperate rocky coastal hydrothermal vent system (Whakaari-White Island, Bay of Plenty, New Zealand): Constraints for ocean acidification studies. Mar. Freshw. Res. **71**: 321–344. doi:10.1071/ MF19167

Acknowledgments

We thank A. Savona and P. Scariano for field activities and C. D. Tramati for stable isotope analyses. This study was carried out within the RETURN Extended Partnership and received funding from the European Union - NextGenerationEU through the Italian Ministry of University and Research under the National Recovery and Resilience Plan NRRP - Mission 4, Component 2, Investment 1.3 - D.D. 1243 8 February 2022, PE0000005 - CUP: B73C22001220006. The views and opinions expressed are those of the authors only and do not necessarily reflect those of the European Union or the European Commission. Neither the European Union nor the European Commission can be held responsible for them. This research was also supported by University of Palermo (FFR).

Submitted 27 June 2023 Revised 30 October 2023 Accepted 08 November 2023